Author's response to reviews

Title: Risk of human papillomavirus-related cancers among patients with end-stage renal disease - an observational cohort study

Authors:

Lars Skov Dalgaard (larsdalg@rm.dk)
Ulrik Fassel (ulrikfassel@gmail.com)
Lars J Østergaard (larsoest@rm.dk)
Bente Jespersen (biesper@dadlnet.dk)
Ole Schmeltz Søgaard (olesoega@rm.dk)
Søren Jensen-Fangel (soejense@rm.dk)

Version: 7 Date: 12 June 2013

Author's response to reviews: see over
Dear Editors at BMC Nephrology

Thank you very much for your comments on our manuscript. Please find the comment and answer enclosed below in this document. A revised manuscript has also been submitted.

Word count for the abstract: 255 and for the revised manuscript: 3548

Please, do not hesitate to contact us if you have any other question or further explanations are needed.

Sincerely,
Lars Skov Dalgaard

Editorial points:
Minor essential revision

Background, first para: “Oncogene” should read “Oncogenic”.

Answer: We have changed “Oncogene” to “Oncogenic”